Author’s response to reviews

Title: Rural-Urban and Racial-Ethnic Differences in Awareness of Direct-to-Consumer Genetic Testing

Authors:
Ramzi Salloum (rsalloum@ufl.edu)
Thomas George (aaa@ufl.edu)
Natalie Silver (aaaa@ufl.edu)
Merry-Jennifer Markham (aaaaa@ufl.edu)
Jaclyn Hall (aaaaaaa@ufl.edu)
Yi Guo (bbb@ufl.edu)
Jiang Bian (bbbb@ufl.edu)
Elizabeth Shenkman (bbbb@ufl.edu)

Version: 1 Date: 11 Oct 2017

Author’s response to reviews:

We have revised the manuscript to include the following:

1. Ethics declaration
2. Email addresses on title page.
3. Conclusions section
4. Acknowledgements
5. Author contributions.